Author’s response to reviews

Title: Readmission rates in non-profit vs. proprietary hospitals before and after the Hospital Readmission Reduction Program Implementation

Authors:

Lauren Birmingham (lbirming@kent.edu)
Willie Oglesby (Billy.Oglesby@jefferson.edu)

Version: 2 Date: 11 Jan 2018

Author’s response to reviews:

Dear Mr. Williams,

Thank you for the opportunity to revise our manuscript. I have responded to each comment below.

Sincerely,

Lauren Birmingham

1. There are some reviewer comments which remain - see end of email. Please address them and amend your manuscript accordingly.

** My co-author and I have re-read the manuscript in its entirety and have made several edits to address run-on sentences, etc. Thank you for this feedback.

2. Please include statements on Ethics and Consent to Participate in one single section.

**I have combined the Ethics and Consent to Participate into one section (line 338).

3. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors
Please also provide more information on the contributions of each author, as per our submission guidelines - https://bmchealthservres.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

**I have made the requested revisions to this section (line 346).**

4. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

**I added email addresses both both authors on the title page. The corresponding author is still designated.**

5. Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. Figure files should contain only the image/graphic, as well as any associated keys/annotations. Where titles/legends are present within the figure files, please remove them.

**I added "Figure Legends" after the references. I removed the title from the single figure that is attached as a separate document.**

6. Please remove any additional files you do not wish to be published alongside your manuscript.

**Only the manuscript and the figure is included in the file inventory.**

7. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

**Only clean copies of the manuscript and figure have been included.**